Author's response to reviews

Title: What factors explain the number of physical therapy treatment sessions in patients referred with low back pain; a multilevel analysis

Authors:
Ilse CS Swinkels (i.swinkels@nivel.nl)
Raymond H Wimmers (r.wimmers@hartstichting.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)
Wil JH van den Bosch (w.vandenbosch@haq.umcn.nl)
Joost Dekker (j.dekker@vumc.nl)
Cornelia HM van den Ende (e.vandenende@cukz.umcn.nl)

Version: 3 Date: 10 October 2005

Author's response to reviews:

Dear editorial team,

Thank you for your decision to accept our manuscript in principle. It is our pleasure to re-submit our manuscript which is improved in response to the final comments made by the reviewers.

We rephrased 'physical therapy utilization' into 'number of physical therapy treatment sessions' in the first sentence of the abstract and background sections. Furthermore, the manuscript is once again checked by an English-language reviser.

In the assumption that we have satisfactorily addressed the points raised by the reviewers, we look forward to hearing from you.

Sincerely,
On behalf of all authors,

Ilse Swinkels